Author's response to reviews

Title: Depression and anxiety, an Indicated Prevention (DIP) protocol in homes for the elderly: feasibility and (cost) effectiveness of a stepped care programme [ISCRTN27540731]

Authors:

  Els E Dozeman (els.dozeman@vumc.nl)
  Anneke DJF van Schaik (a.van.schaik@ggzba.nl)
  Aartjan ATF Beekman (aartjanb@ggzba.nl)
  Wim WAB Stalman (w.stalman@vumc.nl)
  Judith JE Bosmans (judith.bosmans@falw.vu.nl)
  Harm HWJ van Marwijk (hwj.vanmarwik@vumc.nl)

Version: 2 Date: 19 February 2007

Author's response to reviews:

Inclusion and exclusion are slightly revisited:
Residents meeting criteria for major depression and/or clinical anxiety disorder according to the Mini International Neuropsychiatric Interview (MINI) [31,32] or suffering cognitive impairment according to the Mini Mental State Examination (MMSE <21) are excluded. Residents capable of informed consent and with sufficient understanding of the Dutch language, are eligible for the study.

Furthermore:
trial registration number, competing interests and authors' contributions are added.